
Daily Status Tues. 27-Nov-12
Status:
All stations reporting in am

On-Site: John

Weather: Clearing, fairly calm overnight.   Roughly 40 cirrus cover in am.   Chilly, temps dropped to -8 to -10 since roughly 10-11pm last night.   Winds 
fairly light roughly <2 from west.    No evidence of fog, dew, but very light frost with humidity rather high, 80%ish.   

Gully:  Pocketec local storage swapped.   Out #11 with data from 11/21 through 26th.   In #8

Spare DSM gray box (unlabeled) returning to Boulder for misc. testing support there.

Sensors:

Sensors:  Looking good.   A few diag hits on ec150 sonic "low signal" but data ok.

Radiometers, Licors Cleaned: 8:55-9:05 MST,  with methanol.   Frost was on uplooking Rsw but didn't appear to be any on Rlw.    Rsw.both desiccant was 
shot: replaced 9:40-9:50MST.
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